Author's response to reviews

Title: Involving users in the design of a randomised controlled trial of an intervention to promote early presentation in breast cancer: qualitative study

Authors:

Lindsay JL Forbes (lindsay.forbes@kcl.ac.uk)
Carol McNaughton Nicholls (Carol.McnaughtonNicholls@natcen.ac.uk)
Louise Linsell (Louise.linsell@npeu.ox.ac.uk)
Jenny Graham (Jenny.graham@natcen.ac.uk)
Charlotte Tompkins (Charlotte.tompkins@natcen.ac.uk)
Amanda J Ramirez (amanda-jane.ramirez@kcl.ac.uk)

Version: 4 Date: 20 December 2010

Author's response to reviews:

Dear Editor
As requested I have:
1. changed the author list
2. cited the references correctly
3. added UK in the affiliations
4. put the names and email address in separate sections
5. renamed the Background section
6. removed all highlighting and tracking
7. included full details of reference 2
8. corrected a few typos and minor corrections

Best wishes
Lindsay